Author's response to reviews

Title: Controversies in Neurology: Why Monoamine Oxidase B Inhibitors Could Be a Good Choice for the Initial Treatment of Parkinson's Disease

Authors:

Matthias Löhle (matthias.loehle@uniklinikum-dresden.de)
Heinz Reichmann (heinz.reichmann@uniklinikum-dresden.de)

Version: 4 Date: 16 September 2011

Author's response to reviews: see over
Dear editors of BMC Neurology,

attached please find the corrected final version of our manuscript entitled:

“Controversies in Neurology: Why Monoamine Oxidase B Inhibitors Could Be a Good Choice for the Initial Treatment of Parkinson’s Disease”.

We have made the following formatting changes to make the manuscript consistent with BMC neurology:

1) Title: The title is identical in the manuscript and the submission system.

2) Structure: The manuscript now has the correct structure using these headings: Abstract; Background; Discussion; Summary; Competing interests; Authors' contributions; References.

3) Author list: Titles have been removed.

4) Structure: Introduction was changed to background.

5) Typography: We have checked the typography.

We hope the article now meets the requirements to be published in BMC Neurology.

Thank you and best wishes,

Matthias Loehle, MD